Author's response to reviews

Title: Association of physical activity with hypertension and diabetes mellitus among Malaysian adults: a cross-sectional study

Authors:

Chien Huey Teh (chienhuey@imr.gov.my)
Ying Ying Chan (chan.yy@moh.gov.my)
Kuang Hock Lim (limkh@moh.gov.my)
Chee Cheong Kee (kee@imr.gov.my)
Kuang Kuay Lim (limkk@moh.gov.my)
Pei Sien Yeo (yeops@moh.gov.my)
Omar Azahadi (drazahadi@moh.gov.my)
Yusoff Fadhli (fadhli_my@moh.gov.my)
Aris Tahir (tahir.a@moh.gov.my)
Han Lim Lee (leehl@imr.gov.my)
Wasi Ahmad Nazni (nazni@imr.gov.my)

Version: 2 Date: 24 September 2014

Author's response to reviews: see over
COVER LETTER

To the Editors of *BMC Public Health*:

Few changes had been made in this revised version according to your comments:

1. Line and page numbers have been included into the manuscript as per the Editorial Office's comments.

2. "Authors' Contributions" section had been revised according to the preferred format (i.e. used initials to refer to authors)

Your sincerely,

Teh Chien Huey (Corresponding author)

Institute for Medical Research
Ministry of Health Malaysia
Jalan Pahang, 50588 Kuala Lumpur
Malaysia.

Tel:  +603-2616 2689  
HP:  019-3339 505  
Fax: +603-2616 2689  
Email: chienhuey@imr.gov.my  
chienhuey@gmail.com